A 640-Gb/s 4 × 4-MIMO D-Band CMOS Transceiver Chipset

Chenxin Liu[®], Graduate Student Member, IEEE, Zheng Li[®], Member, IEEE,

Yudai Yamazaki[®], Graduate Student Member, IEEE, Hans Herdian[®], Member, IEEE,

Chun Wang¹⁰, Member, IEEE, Anyi Tian¹⁰, Jun Sakamaki, Han Nie¹⁰, Xi Fu¹⁰, Member, IEEE,

Sena Kato^(D), Graduate Student Member, IEEE, Wenqian Wang, Graduate Student Member, IEEE,

Hongye Huang[®], Member, IEEE, Minzhe Tang, Graduate Student Member, IEEE,

Dingxin Xu^D, Member, IEEE, Shinsuke Hara^D, Member, IEEE, Akifumi Kasamatsu^D, Member, IEEE,

Takashi Tomura, Member, IEEE, Hiroyuki Sakai¹⁰, Senior Member, IEEE,

Kazuaki Kunihiro[®], Member, IEEE, Atsushi Shirane, Member, IEEE,

and Kenichi Okada¹⁰, *Fellow*, *IEEE*

Abstract—This work presents a D-band (110-170 GHz) CMOS transceiver (TRX) chipset that covers a 56-GHz (114-170 GHz) signal-chain bandwidth. Both the transmitter (TX) and the receiver (RX) operate as heterodyne architectures with external intermediate frequency (IF) and local oscillator (LO) signals. An eight-way low-Q power-combined power amplifier (PA), a two-way low-Q power-combined low-noise amplifier (LNA), wideband-impedance-transformation passive mixers, commonsource-based cascaded distributed amplifiers (DAs), and a low-loss wideband chip-to-waveguide printed circuit board (PCB) transition are proposed with improved bandwidth and linearity to support high-order wideband quadrature amplitude modulation (QAM) signals. The TRX chipset was fabricated using a 65-nm CMOS process. The TX achieves a 13-dBm saturated output power at 130 GHz with 1150-mW dc power. The RX achieves a 12-dB noise figure (NF) with 550-mW dc power. The proposed TRX chipset achieves a data rate of 200 Gb/s by 32QAM in the single-input-single-output (SISO) over-the-air (OTA) measurement at 0.32 m. A data rate of 150 Gb/s by 16QAM is realized with 43-dBi antennas at 15 m. Additionally, a 640-Gb/s 4 × 4 OTA line-of-sight multiinput multioutput (LOS-MIMO) is demonstrated at 0.52 m.

Index Terms—200 Gb/s, 640 Gb/s, 6G, CMOS, D-band, line-of-sight multiinput multioutput (LOS-MIMO), single-input-

Received 23 August 2024; revised 31 October 2024; accepted 2 December 2024. This article was approved by Associate Editor Ron Kapusta. This work was supported in part by the Ministry of Internal Affairs and Communications (MIC) under Grant JPJ000254; in part by the National Institute of Information and Communications Technology (NICT) under Grant JPJ012368C00801; in part by the Japan Society for the Promotion of Science (JSPS) under Grant JP20H00236; in part by the Special Award for Tokyo Tech Advanced Researchers (STAR); in part by VLSI Design and Education Center (VDEC) in collaboration with Cadence Design Systems, Inc.; in part by Mentor Graphics, Inc.; and in part by Keysight Technologies Japan, Ltd. (*Corresponding author: Chenxin Liu.*)

Chenxin Liu, Zheng Li, Yudai Yamazaki, Hans Herdian, Chun Wang, Anyi Tian, Jun Sakamaki, Han Nie, Xi Fu, Sena Kato, Wenqian Wang, Hongye Huang, Minzhe Tang, Dingxin Xu, Takashi Tomura, Hiroyuki Sakai, Kazuaki Kunihiro, Atsushi Shirane, and Kenichi Okada are with the Department of Electrical and Electronic Engineering, Institute of Science Tokyo, Tokyo 152-8550, Japan (e-mail: liucx@ssc.pe.titech.ac.jp).

Shinsuke Hara and Akifumi Kasamatsu are with the National Institute of Information and Communications Technology, Tokyo 184-8795, Japan.

Color versions of one or more figures in this article are available at https://doi.org/10.1109/JSSC.2024.3515640.

Digital Object Identifier 10.1109/JSSC.2024.3515640

single-output (SISO), subterahertz (sub-THz), transceiver (TRX) chipset, wideband design.

1

I. INTRODUCTION

THE ever-changing technological advances and social developments are placing increasingly higher demands on the throughput of wireless communications [1], [2], [3], [4], [5]. Recent new applications such as smart cars and high-speed indoor devices require data rates at hundreds of Gb/s levels [6]. Over the past decade, millimeter-wave (mm-Wave) and subterahertz (sub-THz) bands have been emerging as promising candidates for the 5G and 6G of high-speed wireless communications due to their high bandwidth capability [7], [8], [9], [10], [11], [12], [13], [14], [15], [16], [17], [18], [19], [20], [21].

The reported wireless single-input–single-output (SISO) data rates with a bit error rate (BER) smaller than 10^{-3} in the past 20 years keep increasing. It could be seen that several wireless communication systems with data rates exceeding 100 Gb/s have been reported, demonstrating the potential and feasibility of next-generation high-speed wireless communication systems [21], [22], [23], [24], [25], [26], [27], [28], [29], [30]. However, such reported previous works operating at the sub-THz frequency band may suffer from low output power, limited bandwidth, and insufficient signal-to-noise ratio (SNR), which limit the wireless data rate and communication distance [19], [21], [23], [27], [29].

Considering the recent development trend [7], [8], [9], [10], [11], [12], [13], [14], [15], [16], [17], [18], [19], [20], [21], [22], [23], [24], [25], [26], [27], [28], [29], [30], it is reasonable to set a higher target for wireless communication speed and distance compared with other state-of-the-art works. To achieve such kinds of targets, a D-band CMOS transceiver (TRX) chipset that covers a 56-GHz (114–170 GHz) signal-chain bandwidth is proposed in this work. The proposed TRX chipset includes wideband and high-linearity techniques that could support wideband high-order quadrature amplitude modulation (QAM) signals. In the SISO over-the-air

© 2024 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

IEEE JOURNAL OF SOLID-STATE CIRCUITS



Fig. 1. Block diagram of the proposed D-band CMOS TRX chipset.

(OTA) measurements, the TRX chipset achieves a data rate of 200 Gb/s by 32QAM at a distance of 0.32 m. A data rate of 150 Gb/s by 16QAM is realized with 43-dBi antennas at a distance of 15 m. To further increase the wireless throughput, a 4 \times 4 line-of-sight multiinput multioutput (LOS-MIMO) structure with four sets of TRX modules is also demonstrated with a total wireless throughput of 640 Gb/s at a distance of 0.52 m. All the measured results satisfy the error vector magnitude (EVM) rms requirements for BER < 10⁻³ [31]. To the best of the author's knowledge, this work realizes the highest SISO and MIMO wireless data rate as of June 2024 compared with other reported works.

The rest of this article is organized as follows: Section II explains the TRX chipset system configuration, system link budget calculation, and LOS-MIMO setup. Section III shows the details of circuit block designs. Section IV describes the printed circuit board (PCB) interface designs. Section V demonstrates the measurement setups and results. Section VI concludes this article.

II. SYSTEM CONFIGURATION

A. Overall System Configuration

Fig. 1 shows the block diagram of the TRX chipset and its frequency allocation. Both the transmitter (TX) and the receiver (RX) are heterodyne architectures with external local oscillator (LO) signals.

The proposed TX consists of a quasidifferential distributed amplifier (DA), a passive double-balanced upconversion mixer, an eight-way power-combined power amplifier (PA), a frequency quadrupler with LO amplifiers, and a serial peripheral interface (SPI) controller. The quasidifferential DA using cascaded topology with single-stage amplifiers provides 56-GHz bandwidth for intermediate-frequency (IF) signals from 4 to 60 GHz. The double-balanced mixer supports dc to 60 GHz at the IF port with theoretical all-ports isolation, greatly reducing the LO leakage to the radio frequency (RF) port [32]. A 90–180-GHz Marchand balun is placed at the output of the mixer to combine the differential RF signals. A high-pass filter (HPF) is placed after the balun to reduce unwanted frequency components. The PA combines eight ways using transmission lines (TLs) to improve the output power, and it has a 56-GHz RF bandwidth from 114 to 170 GHz.

The proposed RX includes a two-way power combined low-noise amplifier (LNA), a passive single-balanced downconversion mixer, a single-ended DA, a frequency quadrupler with LO amplifiers, and an SPI controller. The two-way powercombined LNA utilizes a similar combiner with the PA to improve LNA linearity for wideband signals. It also provides the same RF bandwidth from 114 to 170 GHz as the PA. The single-balanced mixer reduces $LO_{1/4}$ harmonics leakage to the IF port, and it also improves conversion gain (CG) and linearity by using differential LO signals [32]. The same balun with the TX is placed at the mixer input port to generate differential signals. The DA in the RX shares the same design as the TX, and the TX and RX chips also use the same LO chain design. The 25–30-GHz external $LO_{1/4}$ signal's frequency is quadrupled inside the chip, and a differential LO signal is generated to drive the mixers. The differential LO amplifiers use the capacitive-cross-coupled neutralization architecture to increase the gain and reduce the differential mismatch of the LO signals [33]. The biases and switches are controlled through the SPI controller in both the TX and the RX.

To keep the wideband and low-loss characteristics of the connection between the RF ports of the TRX chipset and antennas, which is essential for the communication distance and speed, a PCB transition from the CMOS chip to the standard waveguide flange (UG-387) is included in both the TX and RX modules. It can cover an RF bandwidth from 112 to 170 GHz with a flat frequency response.

In general, high-order QAM SISO signals with data rates up to 200 Gb/s are supported in the entire signal chain with the proposed TRX chipset.

LIU et al.: 640-Gb/s 4 \times 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Fig. 2. Link budget calculations of the proposed TRX chipset. (a) Power-level diagram of the TRX. (b) Calculated TX-to-RX SNDR versus input power at 1 m. (c) Calculated TX-to-RX SNDR versus communication distance with $P_{IN} = -25$ dBm, $P_{IN} = -20$ dBm, and $P_{IN} = -15$ dBm. (d) Simulated group delay of the overall link. (e) Degradation of the mixer SNR due to the LO phase noise.

TABLE I REQUIRED SNR FOR BER $< 10^{-3}$

Modulation	BPSK	QPSK	16QAM	32QAM	64QAM	128QAM	256QAM
SNR (MER) [dB]	5.16	9.80	16.54	19.58	22.55	25.49	28.41
EVM _{RMS} [%]	55.21	32.36	14.89	10.50	7.46	5.32	3.80
EVM _{RMS} [dB]	-5.16	-9.80	-16.54	-19.58	-22.55	-25.49	-28.41

B. TX-to-RX Link Budget Calculation

To evaluate the effective data rate, a BER threshold value is selected as 10^{-3} to maintain a sufficient coding rate after the forward error correction (FEC) [14], [21], [24], [29]. For the *M*-order QAM signal, the required SNR can be calculated by the following equation [34]:

$$SNR = \log_2 M \frac{M - 1}{3 \log_2 \sqrt{M}} \left(\operatorname{erfc}^{-1} \left(\frac{\operatorname{BER} \cdot \log_2 \sqrt{M}}{1 - \frac{1}{\sqrt{M}}} \right) \right)^2$$
(1)

where all variables are in plain ratios. In the measurement, EVM_{rms} values are used for the BER evaluation. The relationship between SNR, EVM, and modulation error ratio (MER) is explained in the following equation:

$$SNR[dB] = MER[dB] = -20 lg(EVM_{rms}[\%]/100).$$
 (2)

Table I summarizes the required SNR (MER) and EVM_{rms} (not EVM_{MAX}) [31].

To get BER $< 10^{-3}$, the SNR at the final output port must be larger than the values of Table I. In real communication systems, nonlinearity also needs to be considered, so that the signal-to-noise-and-distortion ratio (SNDR) is used here to estimate system performance. The SNDR is defined as the ratio between signal power and the sum of noise and intermodulation distortion (IM3) [24].

Fig. 2(a) illustrates the link budget calculations, based on the simulation results of the proposed TRX chipset in the SISO case. The calculation utilizes a 140-GHz signal with a 50-GHz bandwidth at 290 K. The cascaded output 1-dB compression point (OP1dB), signal power level, and noise power level are presented at each stage. The input power level is -20 dBm to make sure that SNDR has sufficient margin. Here, the target modulation is 32QAM. Fig. 2(b) depicts the calculated TX-to-RX SNDR versus the TX input power at 1 m. The TX input power should be larger than -32 dBm and smaller than -18 dBm to maintain an SNDR exceeding 19.58 dB. Fig. 2(c) shows the calculated TX-to-RX SNDR versus the communication distance for various TX input powers of -25, -20, and -15 dBm, respectively. For short-range cases, a lower input power level can be employed to ensure enough SNDR, whereas a higher input power level with binary phase-shift keying (BPSK) or quadrature phaseshift keying (QPSK) can be utilized for long-range scenarios.

The upper calculation assumes a flat group delay and the use of ideal LO signals without any phase noise. It is evident that these two factors exert a considerable influence on the SNDR in actual wireless communications, and thus, they must be considered in the link budget calculations.

The group delay represents the distortion of the modulated signals. Fig. 2(d) shows the simulated group delay of the overall link. A nonflat group delay will cause intersymbol interference (ISI), which makes EVM worse. The degradation

IEEE JOURNAL OF SOLID-STATE CIRCUITS

of the EVM can be calculated by obtaining the phase ripples from the group delay [35]. However, this EVM deterioration resulting from nonflat group delay can be reduced by using an equalization function, thereby minimizing signal distortion and enhancing the EVM performance [36]. This is utilized in the measurement mentioned in Section V.

Additionally, nonideal LO signals invariably contain phase noise, which results in a deterioration of the SNR via mixers. The external LO_{1/4} signal features an integrated phase noise component, $\Delta \phi_{SG}^2$, which can be calculated by the following equation [37]:

$$\Delta\phi_{\rm SG}^2 = 2 \int_{1\,\rm kHz}^{25\,\rm GHz} L_{\rm SG}(f) df \tag{3}$$

where $L_{SG}(f)$ is the single sideband (SSB) phase noise of the external LO_{1/4} signal. Therefore, the LO chain will output an integrated phase noise, $\Delta \phi_{LOChain}^2$, whose value is dependent on $\Delta \phi_{SG}^2$, frequency quadrupler, and amplifiers. This value is added directly to the output by the mixers while maintaining the same level [37]. If the input of the mixer has a certain SNR value, SNR_{IN}, the SNR degradation due to the LO phase noise can be calculated by the following equation [38], [39]:

$$\frac{1}{\text{SNR}_{\text{OUT}}} = \frac{E_m^2 + E_\phi^2}{R^2} = \frac{1}{\text{SNR}_{\text{IN}}} + \Delta \phi_{\text{LOChain}}^2 \qquad (4)$$

where *R* represents the reference vector, E_m is the magnitude error vector, and E_{ϕ} is the phase error vector. Fig. 2(e) shows the influence of the LO phase noise, as explained in (4).

The revised link budget considering the phase noise of the LO signal is also illustrated in Fig. 2(a) and (b). With the inclusion of LO phase noise in the calculations, the overall link SNDR exhibited a reduction of approximately 0.5 dB when TX $P_{\rm in} = -20$ dBm.

C. 4 × 4 LOS-MIMO Configuration

Attaining a higher data rate with an SISO system may prove to be challenging, particularly considering the necessity to guarantee both a more expansive bandwidth and a higher SNDR. In contrast, MIMO systems can provide higher data rate capacity without the necessity of increasing bandwidth requirements. LOS-MIMO systems are particularly well-suited for point-to-point wireless communications, offering better spectrum efficiency [40].

Fig. 3 provides an overview of the D-band 4×4 LOS-MIMO with the singular value decomposition (SVD) architecture [41]. The entire LOS-MIMO system is composed of four TX/RX elements, a matrix V, and a matrix U*. The RF signals contain four streams occupying the same frequency band, which means that f_{RF1} , f_{RF2} , f_{RF3} , and f_{RF4} are the same. In this instance, the SVD is applied to the TX–channel–RX matrix

$$\mathbf{R} \cdot \mathbf{H} \cdot \mathbf{T} = \mathbf{U} \cdot \boldsymbol{\Sigma} \cdot \mathbf{V}^* \tag{5}$$

where **H** represents the channel matrix. **T** and **R** are the diagonal matrices that represent the TX and the RX, respectively. The matrices **U** and **V**^{*} are unitary, and Σ is the diagonal matrix with singular values equal to the square root of the



RF₁₋₄: Same RF frequency band up to 50GHz bandwidth.

Fig. 3. Overview of the D-band 4×4 LOS-MIMO with the SVD.

eigenvalues of the matrix $\mathbf{RHT} \cdot (\mathbf{RHT})^*$ [36], [37]. If the fourstream input matrix **X** is applied to the system, the resulting final output matrix **Y** is given by

$$\mathbf{Y} = \mathbf{U}^* \cdot \mathbf{R} \cdot \mathbf{H} \cdot \mathbf{T} \cdot \mathbf{V} \cdot \mathbf{X} = \mathbf{\Sigma} \cdot \mathbf{X}$$
$$= \begin{bmatrix} \sqrt{\lambda_1} \mathbf{x}_1 & \cdots & \mathbf{0} \\ \vdots & \ddots & \vdots \\ \mathbf{0} & \cdots & \sqrt{\lambda_4} \mathbf{x}_4 \end{bmatrix}$$
(6)

where $\lambda_1, \lambda_2, \lambda_3$, and λ_4 represent the eigenvalues of the **RHT** (**RHT**)^{*}. This method allows for the decomposition of MIMO signals. In certain instances, the matrix **V** can be designated as the identity matrix **I**, thereby enabling the MIMO signal decomposition to be conducted exclusively at the RX end [41]. Moreover, for the MIMO system with high isolation between different streams, the **RHT** is approximately equal to the identity matrix **I** if no gain and loss are considered, which implies that **Y** = **X**, and the MIMO systems.

In this work, high-directivity H–V-polarized antennas are utilized to achieve high isolation, so that the SNDR analysis with the SISO case is also useful for the MIMO system. The generation and decomposition of the MIMO signals are performed using external equipment.

III. CIRCUIT BLOCK DESIGN

This section presents a comprehensive analysis of the circuit blocks, including the PA, LNA, DA, mixers, and LO chains. The results of the simulations and measurements are also presented.

A. Power-Combined PA and LNA

Fig. 4(a) explains an in-phase TL-based power combining and low-Q wideband impedance matching techniques in PA design. The PA unit of driving stages uses a TL-embedded gain-boosting structure, which could enhance the CMOS transistor's gain performance at D-band frequencies [43].

The embedded structure comprises two principal parameters, namely the lengths of TL_1 and TL_2 , which are employed to regulate the gain response across the D-band frequencies. The simulation results show that longer lengths of TL_1 and

LIU et al.: 640-Gb/s 4 \times 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Fig. 4. Proposed PA design. (a) PA unit designs and the low-Q power combining technique. (b) Passive efficiency of the eight-way combiner. (c) Simulated PA *S*-parameters. (d) TX CG and P_{out} at 137 GHz.

TL₂ yield a peak at a lower frequency. A shorter TL₁ is preferable, as a longer TL₁ results in a reduction in the *K* factor. Furthermore, all TLs within the embedded structure are 50 Ω . Additionally, the simulation is based on the typical–typical (TT) process corner.

The gain-boosting structure incorporates power feedback at its output port, which constrains the output power. The PA output stage utilizes a single common source transistor without any gain-boosting structure to achieve the maximum possible saturated output power.

Both these PA units utilize a low characteristic impedance TL to connect to the power network, which is equivalent to a large capacitance shunted to the ground (GND). This ensures that the single-ended power network has minimal impact on the impedance matching. The D-band transistor model was created using the methods described in [44].

The PA unit has a 20- Ω output impedance Z_{out-PA} , and it can be tuned by the addition of a series of TL. For wideband matching design, the impedance needs to stay inside the low-Q region on the Smith chart [45]. In the eight-way combiner design, four different kinds of TLs (TL1, TL2, TL3, and TL₄) are utilized to achieve the wideband power combining and impedance matching from 20 (PA) to 50 Ω (load). The combiner is designed based on electromagnetic (EM) simulation using high-frequency structure simulator (HFSS), and its layout size is 1×0.155 mm. The impedance trajectory from 110 to 170 GHz is plotted on the Smith chart, which shows that a low-Q matching (Q < 1) is realized in this design. According to the stand-alone back-to-back combiner test element group (TEG) measurement, a 70% average passive efficiency is achieved over the whole D-band as shown in Fig. 4(b), which is calculated by the following equation:

$$\eta_{\text{passive}} = 10^{(0.5 \times S21)/10} \times 100\% \tag{7}$$

where *S*21 (dB) represents the de-embedded insertion loss of the back-to-back eight-way combiner. It is assumed that a low return loss is present in (7).

Fig. 4(c) shows the simulated S-parameters of the proposed PA. The PA has a 25-dB gain with S11 and S22 values smaller than -10 dB across the 114–170-GHz frequency range. Additionally, a Monte-Carlo simulation for different process corners is performed to see the variation of the PA's gain, which is from 21 to 28 dB in the most unfavorable situations. The measured TX saturated output power at 137 GHz is 13 dBm, while the saturated power of a PA unit is 5.5 dBm, which is shown in Fig. 4(d). The reasons for the discrepancies in Fig. 4(d) will be discussed in Section V-A.

The LNA shares the same amplifier unit and combines technique with PA to enhance the OP1dB. Fig. 5(a) explains the LNA's topology and the EM 3-D model of the combiner. The combiner size is 0.25×0.175 mm. Fig. 5(b) shows the simulated LNA S-parameters. The LNA has an 18-dB gain with S11 and S22 smaller than -10 dB across the same frequency range as PA. A Monte-Carlo simulation incorporating various process corners is also conducted to assess the impact of transistor mismatch. The result shows a gain variation of 15–19.2 dB in the worst case.

Fig. 5(c) shows an average 9-dB noise figure (NF) in the simulation. Fig. 5(d) illustrates the simulated gain and P_{out} as a function of the LNA input power. The realization of a -14.5-dBm input 1-dB point through the two-way power combining is also demonstrated.

B. Cascaded Common-Source-Based DA

Fig. 6(a) illustrates the design of the DA with a bandwidth of 4–60 GHz. The conventional DA can achieve a wide



Fig. 5. Proposed LNA design. (a) LNA schematic. (b) Simulated LNA *S*-parameters. (c) Simulated LNA SSB NF. (d) LNA gain and P_{out} at 140 GHz.

bandwidth but suffers from lower gain due to the additive gain mechanism [46], [47]. In this work, a common-sourcebased cascaded DA is utilized to maintain high gain, low NF, and high linearity at frequencies exceeding 60 GHz [47]. The first (DA₁) and the third (DA₂) stages are designed as conventional DAs with 40- μ m/60-nm transistors. The TLs are tuned to achieve optimal gain flatness. It can be observed that the TLs on the output side of the DA₁ and DA₂ are longer than those on the input side. This is due to the presence of parasitic capacitance to the ground associated with C_{1-4} , which results in a larger delay. The second stage consists of two cascaded single-stage amplifiers with 80- μ m/60-nm transistors, which serve to enhance the DA's gain.

Fig. 6(b) shows the DA S-parameter's simulation and measurement results. An average gain of 18 dB is achieved over the 4–60-GHz frequency range.

Fig. 6(c) shows a 7–8-dB NF in both the simulation and measurement. The measurement is up to 40 GHz due to the equipment limitations.

Fig. 6(d) shows the group delay of the DA. The precision of the measurement above 40 GHz is influenced by the calibration inaccuracies of the vector network analyzer (VNA) employed in the measurement process.

Fig. 6(e) demonstrates a 10-dBm saturated output power at 30 GHz. In comparison to previous studies [46], [47], the supply voltage is reduced to 1 V with the use of common-source transistors.

C. Wideband Passive TX/RX Mixers

Fig. 7 presents an explanation of TX/RX mixers, TX HPF, and the differential LO driver designs, accompanied by simulation results of mixer CG.

The TX mixer is designed as a double-balanced passive mixer with theoretical all-ports isolation. The differential RF signals at the mixer output are combined by a 1.2-dB-loss Marchand balun with a frequency range of 90–180 GHz [48]. The IF port is a differential port that is connected to the output of the DAs. The LO port is a single-ended port that



Fig. 6. Proposed DA design. (a) DA's schematic. (b) DA S-parameters. (c) DA NF. (d) DA group delay. (e) DA CG and P_{out} .

is followed by a balun and a differential LO driver. The differential capacitive-cross-coupled LO driver is designed to reduce the mismatch of differential LO signals [33], providing a saturated LO power of 8 dBm to the gate of the switch transistors. The measured level of leakage from LO to RF is -40 dBm. An HPF with a 3-dB attenuation difference at 102 GHz is positioned after the RF balun. A 40-dB attenuation difference for the image band can be achieved in combination with the PA and PCB transition. The RX mixer is designed as a single-balanced passive mixer. It has the same LO driver as the TX mixer. The differential RF signal is generated by the same balun design as the TX. RF signals are mixed with differential LO signals, which can enhance the CG in comparison to a single-switch mixer [32]. A single-ended IF signal is generated for external equipment. In the RX on-wafer measurement utilizing RF probes, no LO_{1/4} harmonic leakage is observed

LIU et al.: 640-Gb/s 4 \times 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Fig. 7. Proposed mixer designs. (a) TX double-balanced mixer, RX single-balanced mixer, and differential LO driver designs. (b) TX mixer's CG with swept switch size: 2 μ m/60 nm, 6 μ m/60 nm, 12 μ m/60 nm, 20 μ m/60 nm, $f_{LO} = 110$ GHz. (c) RX mixer's CG with swept switch size: 8 μ m/60 nm, 16 μ m/60 nm, 24 μ m/60 nm, 32 μ m/60 nm, and $f_{LO} = 110$ GHz.

at the IF output. The LO_{1/4} harmonic leakage through the pad (PAD), substrate, and power network is significantly attenuated by distributed GND cells and decoupling capacitors connected to the substrate and power network, respectively. To achieve the maximum CG, both the TX mixer and the RX mixer utilize bias voltage close to $V_{\rm th}$ (0.3 V) on the gate.

In the case of passive mixers, the input impedance can be considered to comprise two components. The first component is the impedance of the switch, which is composed of R_{SW} and C_{SW} . R_{SW} represents the channel resistance when the switch is on. C_{SW} represents the switch's equivalent capacitance when the switch is off. Furthermore, the load impedance is converted to the input frequency with a specific factor, which constitutes the second part [49]. The mixer equivalent circuits are illustrated in Fig. 7(a). In such cases, a larger switch size may provide a higher CG at lower frequencies, but this is accomplished by a gain drop at higher frequencies due to a larger C_{SW} . Conversely, a smaller switch size may offer less C_{SW} , but the CG is constrained across the entire band due to a larger R_{SW} .

To achieve optimal CG and gain flatness over the entire D-band frequency range, it is necessary to select an appropriate switch size to ensure good impedance matching and a small C_{SW} . Fig. 7(b) depicts the TX mixer's CG as a function of the swept switch size. The optimal switch size is determined to be $6 \,\mu$ m/60 nm (black dotted line), which enables the attainment of a flat gain characteristic and an average simulated CG of approximately -10 dB over the whole D-band frequency range. Fig. 7(c) shows the CG of the RX mixer with sweeps of the switch size. The optimal switch size is determined to be 16 μ m/60 nm (black dotted line), with an average simulated CG of approximately -10 dB across the entire D-band frequency range. Both the TX and RX mixers use 110-GHz LO here. Furthermore, the CG and bandwidth characteristics of the proposed mixers exhibit negligible variation across different process corners.

It is also noteworthy that a pair of TL shunts are positioned at the RF side of both the TX and RX mixers. The TL shunts provide a signal path for lower IF frequencies and also serve impedance-matching functions for the RF side.

D. LO Chain

Fig. 8(a) demonstrates the circuit design for the LO frequency. The entire LO chain comprises an $LO_{1/4}$ frequency quadrupler and an LO amplifier.

The $LO_{1/4}$ frequency quadrupler is comprised of an input amplifier (Amp. 1), a middle-stage amplifier (Amp. 2), and two frequency doublers. The input amplifier provides input matching from 25 to 30 GHz. The middle-stage amplifier is responsible for amplifying signals in the frequency range of 50-60 GHz. Amp. 2 utilizes a single amplifier to reduce power consumption and reduce layout area. Moreover, the signal power level at 50-60 GHz remains sufficiently high to justify the inclusion of a single amplifier. The frequency doublers utilize common-source transistors with a low bias voltage in proximity to V_{th} (0.3 V) to enhance the second harmonic component [19]. The LO amplifier comprises three stages using a similar gain-boosting architecture as PA units to drive the LO signal to the saturated level. To reduce the unwanted harmonic components, a 90° open-TL shunt at $LO_{3/4}$ is placed, which serves as a ground short for the $LO_{3/4}$ signals. Furthermore, the narrowband characteristics of the final LO amplifier can be used to constrain other harmonics.

Fig. 8(b) shows the simulated input matching of the $LO_{1/4}$ frequency quadrupler. S11 is less than the -10 dB from 25 to 30 GHz. Fig. 8(c) depicts the output power of the $LO_{1/4}$, $LO_{2/4}$, $LO_{3/4}$, and LO signals versus the input power of the external $LO_{1/4}$ signal, with the $LO_{1/4}$ frequency set to 27.5 GHz. The simulation result shows a 35-dB constraint on the maximum unwanted harmonic component in comparison



Fig. 8. Proposed LO chain design. (a) $LO_{1/4}$ frequency quadrupler and LO amplifier designs. (b) Input matching of the LO chain. (c) $LO_{1/4}$, $LO_{2/4}$, $LO_{3/4}$, and LO output power versus the input power when $f_{LO} = 110$ GHz. (d) Simulated SSB phase noise of the LO chain, incorporating the differential LO driver discussed in Section III-C.

to the LO signal. This could be further reduced by the LO driver discussed in the mixer section.

It can be observed that the entire LO chain contributes to the phase noise on the LO signal. The results of the simulation are presented in Fig. 8(d), which demonstrates an approximate 15-dB increase, resulting in an integrated phase noise of -34.7 dBc from 1 kHz to 25 GHz with extrapolations.

IV. PCB INTERFACE DESIGN

A well-designed PCB is a crucial component of the proposed wireless communication system, as it serves as the interface for the IF, LO, and RF signals. In general, a PCB with low loss and a wideband frequency response is preferable.

This section presents a series of PCB designs. The images of the TRX modules and comprehensive flip-chip components are provided, accompanied by a detailed explanation of the PCB transition from the CMOS chip to the standard rectangular waveguide. Additionally, the PCB design of the connector part for IF and LO signals is also demonstrated.

A. TX/RX Modules and Flip-Chip Implementation

Fig. 9 illustrates the TRX modules' photographs and implementations of the TRX chipset.

LO_{1/4} Control Logic TX Chip Detailed TX flip-chip part

IEEE JOURNAL OF SOLID-STATE CIRCUITS



Fig. 9. Photographs of TRX modules. (a) TX module and top-viewed flip-chip implementation. (b) RX module and top-viewed flip-chip implementation.

Fig. 9(a) illustrates the specifics of the TX module. The TX module features an $LO_{1/4}$ signal interface, a differential IF signal interface, and an RF signal interface. Fig. 9(b) illustrates the details of the RX module. Similarly, the RX module is equipped with an $LO_{1/4}$ signal interface, a single-ended IF signal interface, and an RF interface. The $LO_{1/4}$ and IF signal interfaces are based on the Southwest 1892-04A-6 Connectors, which are capable of withstanding frequencies up to 67 GHz. The RF signal interface is a standard waveguide flange (UG-387) to connect with the WR6 (110–170 GHz) waveguide.

The TX and RX PCB modules are fabricated using a material designated as Megtron6, with a PCB size of 60×65 mm. The CMOS chip is implemented on the PCB by the flip-chip technique, with the use of bumps.

B. RF PCB Transition From CMOS Chip to Waveguide

Fig. 10(a) shows PCB information and the proposed RF PCB transition design from the CMOS chip to the WR6 waveguide. The PCB comprises four layers, while the transition is made with the upper three layers, L1, L2, and L3. L1 is the layer that is connected to the CMOS chip and the waveguide, and L3 is the layer that serves as a backshort layer. The height from L1 to L3 is approximately $\lambda/4$ at 140 GHz, which can enhance the power delivery to the waveguide [50], [51]. The entire PCB transition consists of four distinct parts: the coplanar waveguide (CPW), the CPWto-substrate-integrated-waveguide (SIW) part, the SIW, and the SIW-to-WR6 part. The CMOS chip is connected to the PCB CPW through the use of gold bumps. Furthermore, the height of the bumps is optimized at 80 μ m to reduce

LIU et al.: 640-Gb/s 4 \times 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Fig. 10. PCB transition from the CMOS chip to the WR6 waveguide. (a) Transition design and PCB information. (b) Photographs of the proposed transition and customized waveguide. (c) Measured and simulated transition insertion loss over the entire D-band frequency range.

the parasitic capacitance from the signal to the ground. The CPW-to-SIW section comprises a tapered connection, which enables the attainment of a wideband characteristic. The SIW is designed to transmit D-band signals. As for the SIW-to-WR6 transition, a patch on the L1 layer and matching vias with a 75- μ m diameter facilitate an optimized wideband frequency response [50], [51].

Fig. 10(b) illustrates the actual photographs of the PCB transition and the customized waveguide. The top surface of the customized waveguide is utilized for antenna connections, while the bottom side includes an empty aperture for the CMOS chips. Fig. 10(c) demonstrates the simulated *S*-parameters and de-embedded measured insertion loss of the proposed PCB transition. The simulation shows a 3-dB loss and a less than -10-dB average return loss. The measurement result demonstrates a 3–4-dB loss at the frequency range of 112–170 GHz, which provides a wideband RF signal path for the TRX modules.

To provide further evaluation of the performance of design, the proposed PCB transition three different back-to-back PCBs are fabricated and subsequently measured, as illustrated in Fig. 11(a). The first



Fig. 11. Back-to-back PCB transition. (a) Top and bird's eye view of the back-to-back PCB transition design including WR6–SIW–WR6, WR6–CMOS–WR6, and WR6–SPW–WR6. (b) Detailed photographs and measured insertion loss.

back-to-back configuration is a back-to-back WR6–SIW–WR6 structure. Two SIW-to-WR6 transitions are connected through an SIW, and the loss of the SIW-to-WR6 is evaluated. The second back-to-back architecture incorporates a CPW at the center, in addition to the first one, and comprises two CPW-to-SIW transitions, which can be utilized to assess CPW loss. The third back-to-back architecture includes an additional 1.6-mm CMOS TL, based on the second one, to evaluate the loss of the bump connection.

Fig. 11(b) illustrates the detailed photographs of the three back-to-back architectures, along with their simulated and measured insertion loss. It can be observed that the SIW-to-WR6 has a good corresponding result between simulation and measurement. However, the back-to-back architectures, which include CPW and CMOS TL, demonstrate a notable loss increase between simulations and measurement. The observed degradation may be caused by the fabrication mismatch, surface roughness [52], and the potential for inaccurate material parameters at D-band frequency. These measurement results are used to derive the de-embedded PCB transition loss mentioned in Fig. 10(c).

C. Connector Part for IF and LO Signals

Fig. 12(a) shows the PCB design of the connector part for the IF and LO signals. The connector part utilizes the identical PCB configuration as that utilized for the RF transition. The Southwest 1892-04A-6 Connector's signal pin is connected to the PCB CPW on the L1 layer with a 90–150–90- μ m CPW TL. And L3 serves as the CPW's bottom ground. The CPW on



Fig. 12. PCB connections with the Southwest 1892-04A-6 Connector. (a) PCB design of the connector part. (b) 60-mm PCB TL with back-to-back connectors. (c) Simulated insertion loss of the PCB connection with the connector. (d) *S*-parameters of the back-to-back PCB TL of (b).

the PCB has a configuration of 75–90–75 μ m, which results in a 50- Ω characteristic impedance.

Fig. 12(b) shows a photograph of a 60-mm PCB CPW TL with back-to-back connectors, which is utilized for the assessment of the losses associated with the connector part and the PCB TLs.

Fig. 12(c) shows simulated S-parameters of the connector part. A wideband input impedance matching (S11 < -10 dB) is achieved from dc to 60 GHz in the simulation.

Fig. 12(d) shows the measured and simulated *S*-parameters of the back-to-back PCB CPW TL referenced in Fig. 12(b). The measurement results demonstrate that the design realizes a wideband input impedance matching up to 60 GHz, which is in accordance with the simulation results. However, the measurement also depicts an increased attenuation factor compared to the simulation, which may also be attributed to the surface roughness issue, similar to those observed in RF transition designs [52].

V. MEASUREMENT RESULT

This section presents the measurement results of the proposed TRX chipset including TX/RX performance, short-range OTA SISO, long-range OTA SISO, and short-range OTA LOS-MIMO scenarios.

A. TX/RX Modules Performance Measurement

Fig. 13 shows the chip micrographs with a breakdown of block area and power consumption. The proposed TRX chipset



IEEE JOURNAL OF SOLID-STATE CIRCUITS

Fig. 13. TRX chipset. (a) TX micrograph and power consumption. (b) RX micrograph and power consumption.

is fabricated in a 65-nm CMOS process. The TX chip size is 1.87×3.3 mm and the RX chip size is 1.65×2.625 mm. The TX and RX chips consume 1150 and 550 mW with a 1-V power supply and 1468 and 700 mW with a 1.2-V power supply, respectively. In the TX chip, two DAs for the differential IF signals may suffer from the mismatch issue due to their relevant far distance, which can lead to some decrease in CG, accompanied by gain ripples. The discrepancy between the two paths of the differential signals will ultimately result in the summation of a common-mode signal and a differential signal. The common-mode signal will be suppressed by the double-balanced mixer and the RF balun at its output port.

The TX and RX chips are measured individually after being implemented on the PCB. Fig. 14(a) and (b) shows the measurement setup. An external D-band mixer with a bandwidth of 60 GHz for both IF and RF signals is used here to realize frequency up- and downconversion. A Keysight E8257D signal generator (SG) with an external frequency tripler is utilized to provide 110-GHz LO for the external mixer. Another SG provides the LO_{1/4} (27.5 GHz) for the TRX. A Keysight M8194A arbitrary waveform generator (AWG) is used to provide IF signals. A Keysight UXR1104A oscilloscope is

LIU et al.: 640-Gb/s 4 \times 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Fig. 14. TRX measurements. (a) TX module measurement setup. (b) RX module measurement setup. (c) TX CG versus RF frequency. (d) RX CG versus RF frequency. (e) TX saturated output power versus RF frequency with 1/1.2 V PA's drain supply voltage (VDD). (f) RX NF versus RF frequency. (g) Measured TX EVM versus TX output power. (h) Measured RX EVM versus RX input power.

used to demodulate IF signals. Fig. 14(c)-(f) illustrates the TX CG, RX CG, TX P_{sat} , and RX SSB NF, respectively. All results are presented after de-embedding. In the measurement of saturated output power from the TX, the TX chip achieves a 12–15-dBm P_{sat} with a 1.2-V power supply. The results may be limited by the phase mismatch between the differential signals and insufficient output power from the AWG at higher frequencies.

In the CG measurement, both TX and RX chips achieve an 18–20-dB gain from 114 to 170 GHz. The average NF of the RX chip is 12 dB. The discrepancies in TX/RX CG and RX NF gain between the simulation and measurement results are primarily due to the inherent limitations in the accuracy of the passive circuit blocks, the transistor models employed for PA



Fig. 15. Short-range SISO measurement at 0.32 m. (a) Short-range SISO measurement setup with 25-dBi horn antennas. (b) Measurement photograph. (c) Measured EVM and constellations. (d) Measured EVM_{rms} versus the SISO data rate.

units, and the parasitic extraction (PEX) simulation for mixers as well as higher junction temperatures.

Fig. 14(g) and (h) shows the measured EVM results of the TX and RX with 2, 5, and 10 Gbaud 16QAM signals. These results are constrained by the linearity of the external mixer.



Fig. 16. Long-range SISO measurement at 15 m. (a) Long-range SISO measurement setup with 43-dBi antennas and a D-band attenuator. (b) Measurement photograph. (c) Measured EVM and constellations. (d) Measured EVM_{rms} versus the equivalent distance.

To maximize the TX output power, all OTA measurements discussed in the following utilize a 1.2-V power supply for the PA, whereas the other blocks utilize a 1-V supply.

B. Short-Range OTA SISO Measurement

Fig. 15(a) demonstrates the OTA measurement configuration for the short-range SISO scenario. Fig. 15(b) shows the OTA setup's photograph. In this measurement, an AWG (Keysight M8199A) is employed to generate modulated signals up to 60 GHz, and an oscilloscope (Keysight UXR1104A) is utilized to evaluate the constellation and EVM. Two identical SGs (Keysight E8257D) provide external LO_{1/4} (27.5 GHz) signals. The measurement is conducted at a distance of 0.32 m, utilizing the same 25-dBi horn antennas for both the TX and RX modules. The antennas are connected to the proposed



Fig. 17. Short-range LOS-MIMO measurement at 0.52 m. (a) Short-range 4×4 LOS-MIMO measurement setup with 25-dBi antennas. (b) Antenna arrangement and measurement photograph. (c) Measured EVM and constellations. (d) Measured EVM_{rms} versus the data rate of each stream.

RF PCB transition through the standard waveguide flange (UG-387). The maximum data rates, as measured, are 43 Gb/s

	ISSCC 2018 [23]	JSSC 2019 [21]	TMTT 2020 [25]	IMS 2020 [53]	JSSC 2020 [24]	JSSC 2023 [27]	ISSCC 2024 [20]	JSSC 2024 [29]	This work				
Technology	65nm CMOS	40nm CMOS	130nm SiGe	130nm SiGe	80nm InP-HEMT	130nm SiGe	28nm CMOS	65nm CMOS	65nm CMOS				
RF Freq. [GHz]	70 - 105	252 - 279	225 - 255	222.5 - 257.5	275 - 305	110 - 170	90 - 180	88 - 136	114 - 170				
TRX Integration	TRX	TRX	TX & RX	TX & RX	Integrated PA	TRX	TX & RX	TRX	TX & RX				
Modulation*	16QAM	16QAM	QPSK	QPSK / 16QAM	16QAM	BPSK / QPSK 16QAM / 32QAM	оок	16QAM	16QAM	32QAM	64QAM	128QAN	
SISO TX-to-RX Data Rate** [Gb/s]	120	80	55	60 / N/A [#]	120	60 / 120 120 / N/A ^{##}	16 / 20	112	172	200	120	56	
SISO EVM _{RMS} [dB]	-17	-18.42	-9.94	-14.52 / -14.61	-16.7	-11.15 / -9.9 -17.2 / -17.6	N/A	-16.6	-19.23	-19.61	-22.64	-25.57	
MIMO TX-to-RX Data Rate** [Gb/s]	N/A	N/A	110	N/A	N/A	N/A	N/A	N/A	640	N/A	N/A	N/A	
MIMO EVM _{RMS} [dB]	N/A	N/A	-9.94	N/A	N/A	N/A	N/A	N/A	-17.69	N/A	N/A	N/A	
TX P _{sat} [dBm]	-1.9	-1.6	7.5	12	11	-8	15-18	0	12 - 15				
RX SSB NF [dB]	N/A	22.9	17	18.5	10 - 17 ⁺	10.08*	6.2 - 8.5▲	15	12				
TX/RX P _{DC} [mW]	120/160	890/897	2850 (TRX)	1237 / 850	4500 / 4500	2500 / 1950	750 / 160	200 / 120	1150 / 550 (1V) 1468 / 700 (1.2V)				
Antenna Gain [dBi]	23	24	25	21	50	4.8	15	15	25		4	43	
Communication Distance [m]	0.2	0.03	1/2	0.8	9.8	0.15	1 / 0.5	0.02	0.32 / 0.52 15 / 9		5 / 90		
RF Interface	Flip-Chip Implementation	Probing	On-Chip Antenna (Lens)	On-Chip Antenna (Lens)	Individual module	On-Chip Antenna (Lens)	Wire Bonding	Flip-Chip Implementation	Wideband Flip-Chip Implementation & PCB-to-Waveguide Transition				
TX/RX Area [mm²]	6 (TRX)	11.06 (TRX)	4.02/3.36	7 / 5.1	N/A	27.2 (TRX)	0.69 / 0.77 [°]	5.7 (TRX)	6.17 / 4.33				
*This work can also support BPSK & QPSK. [#] Data rate of 100Gb/s is also reported with BER > 10 ³ . *NF of the RF PA. **Data rate for BER < 10 ³ . ^{##} Data rate of 200Gb/s is also reported with BER > 10 ³ . ^A Simulation result.							ିOn-chip PLL						

TABLE II Performance Comparison

in BPSK, 86 Gb/s in QPSK, 172 Gb/s in 16QAM, 200 Gb/s in 32QAM, 120 Gb/s in 64QAM, and 56 Gb/s in 128QAM.

Fig. 15(c) presents a summary of the measured constellations and EVM for BPSK, QPSK, 16QAM, 32QAM, 64QAM, and 128QAM cases. Modulated signals with a single carrier are used here with a roll-off factor of 0.05 (>30 Gbaud) and 0.1 (<30 Gbaud). The measured EVM_{rms} values are -19.26 dB (10.89%, BPSK at 43 Gb/s), -19.8 dB (10.23%, QPSK at 86 Gb/s), -19.23 dB (10.93%, 16QAM at 172 Gb/s), -19.61 dB (10.46%, 32QAM at 200 Gb/s), -22.64 dB (7.38%, 64QAM at 120 Gb/s), and -25.57 dB (5.27%, 128QAM at 56 Gb/s), which all satisfy the 10^{-3} BER criteria discussed in Table I.

Fig. 15(d) depicts the relationship between measured EVM_{rms} values and SISO data rates of various kinds of modulations. The required EVM_{rms} values for 16QAM, 32QAM, 64QAM, and 128QAM are also shown as the black dot lines. The maximum SISO data rate is achieved by 40-Gbaud 32QAM, which is 200 Gb/s.

C. Long-Range OTA SISO Measurement

To further evaluate the TRX modules' performance with extended communication distances, a long-range OTA SISO measurement is also conducted here.

Fig. 16(a) demonstrates the setup for the long-range OTA SISO measurement. Fig. 16(b) shows the measurement photograph. The measurement is conducted inside an anechoic chamber room. Similar to the short-range OTA SISO measurement setup, an AWG (Keysight M8194A) is utilized to generate single carrier modulated signals up to 48 GHz, and

the same oscilloscope (UXR1104A) is utilized for signal evaluation. External LO_{1/4} (27.5 GHz) signals are generated from two SGs (Keysight E8257D). The distance is fixed at 15 m, with two 43-dBi Cassegrain antennas. An additional D-band attenuator is positioned at the TX side to evaluate the equivalent communication distance. The maximum data rates, as measured, are 128 Gb/s in 16QAM at 15 m, 150 Gb/s in 32QAM at 15 m, and 90 Gb/s in 64QAM at 15 m. The maximum equivalent distance for 20-Gb/s 16QAM is 90 m.

Fig. 16(c) summarizes EVM values and constellations for 16QAM, 32QAM, and 64QAM cases. The measured EVM_{rms} values are -16.6 dB (14.79%, 20-Gb/s 16QAM at 90 m), -17.6 dB (13.18%, 128-Gb/s 16QAM at 15 m), -19.89 dB (10.13%, 150-Gb/s 32QAM at 15 m), and -22.62 dB (7.4%, 90-Gb/s 64QAM at 15 m), which all satisfy the 10^{-3} BER criteria.

Fig. 16(d) shows the measured various data rates of 16QAM signals' EVM_{rms} values versus the equivalent communication distance which is tuned by the embedded D-band attenuator at the TX side.

D. Short-Range OTA LOS-MIMO Measurement

Fig. 17(a) shows the setup of the short-range 4×4 LOS-MIMO measurement. Four TRX modules with H–V-polarized 25-dBi antennas form a 4×4 LOS MIMO architecture. The AWG (Keysight M8194A) is used here to generate OFDM MIMO signals up to 48 GHz. And the oscilloscope (Keysight UXR1104A) is utilized for MIMO signal decomposition. Both IF₁₋₄ and RF₁₋₄ signals occupy 44-GHz bandwidth





Fig. 18. Reported SISO data rate with BER $< 10^{-3}$ as of June 2024.

in the frequency band of 4–48 GHz and 114–158 GHz, respectively. Two SGs (Keysight E8257D) deliver external $LO_{1/4}$ (27.5 GHz) signals to the TRX modules with two power dividers. The communication distance is 0.52 m.

Fig. 17(b) shows the antenna arrangement, the measurement photograph, and received power levels of channel 1 (CH1), channel 2 (CH2), channel 3 (CH3), and channel 4 (CH4) when the signal in CH1 is turned on. The antenna pitch is 8 cm. This value represents the minimum that can be achieved given the limitations of the PCB size. Antenna spacing can affect the data rate in the measurement. As discussed previously, a larger spacing makes the MIMO system perform as a combination of four SISO systems. Conversely, small spacing increases the signal power level on the receiving side. This increased power level may result in a more nonlinear system and degradation in EVM, which also limits the data rate.

The measured maximum data rate per stream is 160 Gb/s by the 16QAM OFDM signals. The measured EVM_{rms} values are -17.5 dB (13.34%), -17.94 dB (12.68%), -17.28 dB (13.68%), and -18.05 dB (12.52%) for stream 1 (V-Pol.), stream 2 (H-Pol.), stream 3 (V-Pol.), and stream 4 (H-Pol.), respectively, which satisfies the 10^{-3} BER requirement. The total data rate is 640 Gb/s. The utilization of 16QAM rather than the targeted 32QAM is to achieve the maximum available data rate, considering the constraints imposed by the EVM. Fig. 17(c) shows the summary of the EVM_{rms} values and constellation. Fig. 17(d) shows the relationship of EVM_{rms} values versus the data rate of streams 1-4. It can be observed that there are still some EVM margins from the requirement of the 16QAM signals, which implies the potential of the proposed TRX chipset to achieve a higher data rate with wider bandwidth modulated signals.

VI. CONCLUSION

This work presents a D-band TRX chipset using a 65-nm CMOS process. The TRX chipset is capable of supporting a signal-chain bandwidth of up to 56 GHz. The wideband and improved linearity designs permit the support of high-order wideband QAM signals up to 200 Gb/s in the OTA measurement. A 4×4 LOS-MIMO configuration is also conducted to further increase the total wireless throughput. Fig. 18 illustrates a comparison between this work and other reported SISO data rates with BER $< 10^{-3}$ in the past 20 years.

Table II demonstrates a comparison of the proposed TRX chipset with other reported state-of-the-art wireless TRXs. It shows the highest 200 Gb/s OTA SISO TX-to-RX data rate and the highest 640 Gb/s OTA MIMO TX-to-RX data rate as of June 2024, which provides a potential solution for the next generation of wireless communications.

REFERENCES

- [1] N. Javaid, A. Sher, H. Nasir, and N. Guizani, "Intelligence in IoT-based 5G networks: Opportunities and challenges," *IEEE Commun. Mag.*, vol. 56, no. 10, pp. 94–100, Oct. 2018, doi: 10.1109/MCOM.2018.1800036.
- [2] A. Ghosh, A. Maeder, M. Baker, and D. Chandramouli, "5G evolution: A view on 5G cellular technology beyond 3GPP release 15," *IEEE Access*, vol. 7, pp. 127639–127651, 2019, doi: 10.1109/ACCESS.2019. 2939938.
- [3] L. Chettri and R. Bera, "A comprehensive survey on Internet of Things (IoT) toward 5G wireless systems," *IEEE Internet Things J.*, vol. 7, no. 1, pp. 16–32, Jan. 2020, doi: 10.1109/JIOT.2019.2948888.
- [4] W. Jiang, B. Han, M. A. Habibi, and H. D. Schotten, "The road towards 6G: A comprehensive survey," *IEEE Open J. Commun. Soc.*, vol. 2, pp. 334–366, 2021, doi: 10.1109/OJCOMS.2021.3057679.
- [5] C.-X. Wang et al., "On the road to 6G: Visions, requirements, key technologies and testbeds," *IEEE Commun. Surveys Tuts.*, vol. 25, no. 2, pp. 905–974, 2nd Quart., 2023, doi: 10.1109/COMST.2023.3249835.
- [6] S. Gringeri, E. B. Basch, and T. J. Xia, "Technical considerations for supporting data rates beyond 100 Gb/s," *IEEE Commun. Mag.*, vol. 50, no. 2, pp. s21–s30, Feb. 2012, doi: 10.1109/MCOM.2012.6146482.
- [7] B. Sadhu et al., "A 28-GHz 32-element TRX phased-array IC with concurrent dual-polarized operation and orthogonal phase and gain control for 5G communications," *IEEE J. Solid-State Circuits*, vol. 52, no. 12, pp. 3373–3391, Dec. 2017, doi: 10.1109/JSSC.2017.2766211.
- [8] B. Sadhu et al., "A 24-to-30 GHz 256-element dual-polarized 5G phased array with fast beam-switching support for >30,000 beams," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, vol. 65, San Francisco, CA, USA, Feb. 2022, pp. 436–438, doi: 10.1109/ISSCC42614.2022.9731778.
- [9] J. Pang et al., "A 28-GHz CMOS phased-array beamformer utilizing neutralized bi-directional technique supporting dual-polarized MIMO for 5G NR," *IEEE J. Solid-State Circuits*, vol. 55, no. 9, pp. 2371–2386, Sep. 2020, doi: 10.1109/JSSC.2020.2995039.
- [10] Z. Li et al., "A 39-GHz CMOS bidirectional Doherty phasedarray beamformer using shared-LUT DPD with inter-element mismatch compensation technique for 5G base station," *IEEE J. Solid-State Circuits*, vol. 58, no. 4, pp. 901–914, Apr. 2023, doi: 10.1109/JSSC.2022.3232137.
- [11] Y. Wang et al., "A 39-GHz 64-element phased-array transceiver with built-in phase and amplitude calibrations for large-array 5G NR in 65-nm CMOS," *IEEE J. Solid-State Circuits*, vol. 55, no. 5, pp. 1249–1269, May 2020, doi: 10.1109/JSSC.2020.2980509.
- [12] K. Kibaroglu, M. Sayginer, and G. M. Rebeiz, "A low-cost scalable 32-element 28-GHz phased array transceiver for 5G communication links based on a 2 × 2 beamformer flip-chip unit cell," *IEEE J. Solid-State Circuits*, vol. 53, no. 5, pp. 1260–1274, May 2018, doi: 10.1109/JSSC.2018.2791481.
- [13] S. Mondal and J. Paramesh, "Power-efficient design techniques for mm-wave hybrid/digital FDD/full-duplex MIMO transceivers," *IEEE J. Solid-State Circuits*, vol. 55, no. 8, pp. 2011–2026, Aug. 2020, doi: 10.1109/JSSC.2020.2987691.
- [14] J. Pang et al., "A 50.1-Gb/s 60-GHz CMOS transceiver for IEEE 802.11ay with calibration of LO feedthrough and I/Q imbalance," *IEEE J. Solid-State Circuits*, vol. 54, no. 5, pp. 1375–1390, May 2019.
- [15] M. Fujishima, M. Motoyoshi, K. Katayama, K. Takano, N. Ono, and R. Fujimoto, "98 mW 10 Gbps wireless transceiver chipset with D-band CMOS circuits," *IEEE J. Solid-State Circuits*, vol. 48, no. 10, pp. 2273–2284, Oct. 2013, doi: 10.1109/JSSC.2013.2261192.
- [16] S. Carpenter et al., "A D-band 48-Gbit/s 64-QAM/QPSK directconversion I/Q transceiver chipset," *IEEE Trans. Microw. The*ory Techn., vol. 64, no. 4, pp. 1285–1296, Apr. 2016, doi: 10.1109/TMTT.2016.2533491.

- [17] M. H. Eissa et al., "Wideband 240-GHz transmitter and receiver in BiCMOS technology with 25-Gbit/s data rate," *IEEE J. Solid-State Circuits*, vol. 53, no. 9, pp. 2532–2542, Sep. 2018, doi: 10.1109/JSSC.2018.2839037.
- [18] I. Abdo et al., "A 300 GHz wireless transceiver in 65nm CMOS for IEEE802.15.3d using push-push subharmonic mixer," in *IEEE MTT-S Int. Microw. Symp. Dig.*, Los Angeles, CA, USA, Aug. 2020, pp. 623–626.
- [19] I. Abdo et al., "A bi-directional 300-GHz-band phased-array transceiver in 65-nm CMOS with outphasing transmitting mode and LO emission cancellation," *IEEE J. Solid-State Circuits*, vol. 57, no. 8, pp. 2292–2308, Aug. 2022, doi: 10.1109/JSSC.2022.3179166.
- [20] D. Tang et al., "A 90-to-180 GHz APD-integrated transmitter achieving 18 dBm P_{sat} in 28 nm CMOS," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, San Francisco, CA, USA, Feb. 2024, pp. 411–413, doi: 10.1109/ISSCC49657.2024.10454268.
- [21] S. Lee et al., "An 80-Gb/s 300-GHz-band single-chip CMOS transceiver," *IEEE J. Solid-State Circuits*, vol. 54, no. 12, pp. 3577–3588, Dec. 2019, doi: 10.1109/JSSC.2019.2944855.
- [22] K. Katayama et al., "A 300 GHz CMOS transmitter with 32-QAM 17.5 Gb/s/ch capability over six channels," *IEEE J. Solid-State Circuits*, vol. 51, no. 12, pp. 3037–3048, Dec. 2016, doi: 10.1109/JSSC.2016.2602223.
- [23] K. K. Tokgoz et al., "A 120 Gb/s 16QAM CMOS millimeter-wave wireless transceiver," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, San Francisco, CA, USA, Feb. 2018, pp. 168–170, doi: 10.1109/ISSCC.2018.8310237.
- [24] H. Hamada et al., "300-GHz-band 120-Gb/s wireless front-end based on InP-HEMT PAs and mixers," *IEEE J. Solid-State Circuits*, vol. 55, no. 9, pp. 2316–2335, Sep. 2020, doi: 10.1109/JSSC.2020.3005818.
- [25] P. Rodriguez-Vazquez, J. Grzyb, B. Heinemann, and U. R. Pfeiffer, "A QPSK 110-Gb/s polarization-diversity MIMO wireless link with a 220–255 GHz tunable LO in a SiGe HBT technology," *IEEE Trans. Microw. Theory Techn.*, vol. 68, no. 9, pp. 3834–3851, Sep. 2020, doi: 10.1109/TMTT.2020.2986196.
- [26] S. Callender et al., "A fully integrated 160 Gb/s D-band transmitter with 1.1 pJ/b efficiency in 22 nm FinFET technology," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, vol. 65, San Francisco, CA, USA, Feb. 2022, pp. 78–80, doi: 10.1109/ISSCC42614.2022.9731663.
- [27] A. Karakuzulu, W. A. Ahmad, D. Kissinger, and A. Malignaggi, "A fourchannel bidirectional D-band phased-array transceiver for 200 Gb/s 6G wireless communications in a 130-nm BiCMOS technology," *IEEE J. Solid-State Circuits*, vol. 58, no. 5, pp. 1310–1322, May 2023, doi: 10.1109/JSSC.2022.3232948.
- [28] A. Agrawal et al., "A 128-Gb/s D-band receiver with integrated PLL and ADC achieving 1.95-pJ/b efficiency in 22-nm FinFET," *IEEE J. Solid-State Circuits*, vol. 58, no. 12, pp. 3364–3379, Dec. 2023, doi: 10.1109/JSSC.2023.3315692.
- [29] C. Wang et al., "A sub-THz full-duplex phased-array transceiver with self-interference cancellation and LO feedthrough suppression," *IEEE J. Solid-State Circuits*, vol. 59, no. 4, pp. 978–992, Apr. 2024, doi: 10.1109/JSSC.2024.3353067.
- [30] C. Wang et al., "A 236-to-266 GHz 4-element amplifier-last phased-array transmitter in 65 nm CMOS," in *IEEE Int. Solid-State Circuits Conf.* (*ISSCC*) Dig. Tech. Papers, San Francisco, CA, USA, Feb. 2024, pp. 415–417. [Online]. Available: 10.1109/ISSCC49657.2024. 10454273
- [31] M. Vigilante, E. McCune, and P. Reynaert, "To EVM or two EVMs? An answer to the question," *IEEE Solid-State Circuits Mag.*, vol. 9, no. 3, pp. 36–39, Summer 2017, doi: 10.1109/MSSC.2017.2714398.
- [32] B. Razavi, *RF Microelectronics*. Upper Saddle River, NJ, USA: Prentice-Hall, 1998.
- [33] H. Asada, K. Matsushita, K. Bunsen, K. Okada, and A. Matsuzawa, "A 60 GHz CMOS power amplifier using capacitive cross-coupling neutralization with 16% PAE," in *Proc. 41st Eur. Microw. Conf.*, Manchester, U.K., Oct. 2011, pp. 1115–1118.
- [34] E. McCune, *Practical Digital Wireless Signals*. Cambridge, U.K.: Cambridge Univ. Press, 2010.
- [35] D. Pimingsdorfer et al., "Impact of SAW RF and IF filter characteristics on UMTS transceiver system performance," in *Proc. IEEE Ultrason. Symp.*, vol. 1, Tahoe, NV, USA, Oct. 1999, pp. 365–368, doi: 10.1109/ULTSYM.1999.849420.

- [36] K. Okada et al., "A 60-GHz 16QAM/8PSK/QPSK/BPSK directconversion transceiver for IEEE802.15.3c," *IEEE J. Solid-State Circuits*, vol. 46, no. 12, pp. 2988–3004, Dec. 2011, doi: 10.1109/JSSC.2011.2166184.
- [37] T. Siriburanon et al., "A low-power low-noise mm-wave subsampling PLL using dual-step-mixing ILFD and tail-coupling quadrature injection-locked oscillator for IEEE 802.11ad," *IEEE J. Solid-State Circuits*, vol. 51, no. 5, pp. 1246–1260, May 2016, doi: 10.1109/JSSC.2016.2529004.
- [38] A. Georgiadis, "Gain, phase imbalance, and phase noise effects on error vector magnitude," *IEEE Trans. Veh. Technol.*, vol. 53, no. 2, pp. 443–449, Mar. 2004, doi: 10.1109/TVT.2004.823477.
- [39] K. Okada, "60 GHz WiGig frequency synthesizer using injection locked oscillator," in *Proc. IEEE Radio Freq. Integr. Circuits Symp. (RFIC)*, Jun. 2014, pp. 109–134.
- [40] P. F. Driessen and G. J. Foschini, "On the capacity formula for multiple input-multiple output wireless channels: A geometric interpretation," *IEEE Trans. Commun.*, vol. 47, no. 2, pp. 173–176, Feb. 1999, doi: 10.1109/26.752119.
- [41] T. Maru, M. Kawai, E. Sasaki, and S. Yoshida, "Line-of-sight MIMO transmission for achieving high capacity fixed point microwave radio systems," in *Proc. IEEE Wireless Commun. Netw. Conf.*, Las Vegas, NV, USA, Mar. 2008, pp. 1137–1142, doi: 10.1109/WCNC.2008.205.
- [42] M. D. Larsen and A. L. Swindlehurst, "MIMO SVD-based multiplexing with imperfect channel knowledge," in *Proc. IEEE Int. Conf. Acoust., Speech Signal Process.*, Dallas, TX, USA, Mar. 2010, pp. 3454–3457, doi: 10.1109/ICASSP.2010.5495970.
- [43] H. Bameri and O. Momeni, "A high-gain mm-wave amplifier design: An analytical approach to power gain boosting," *IEEE J. Solid-State Circuits*, vol. 52, no. 2, pp. 357–370, Feb. 2017, doi: 10.1109/JSSC.2016.2626340.
- [44] S. Kawai, S. Sato, S. Maki, K. K. Tokgoz, K. Okada, and A. Matsuzawa, "Accurate transistor modeling by three-parameter pad model for millimeter-wave CMOS circuit design," *IEEE Trans. Microw. Theory Techn.*, vol. 64, no. 6, pp. 1736–1744, Jun. 2016, doi: 10.1109/TMTT.2016.2549527.
- [45] M. Steer, Microwave and RF Design, Volume 3: Networks. Raleigh, NC, USA: North Carolina State University, 2019.
- [46] C.-M. Hsu, Y. Wang, and H. Wang, "A 14–91 GHz distributed amplifier in 65-nm CMOS," in *Proc. IEEE Asia–Pacific Microw. Conf. (APMC)*, Hong Kong, Dec. 2020, pp. 1009–1011, doi: 10.1109/APMC47863.2020.9331580.
- [47] J.-C. Kao, P. Chen, P.-C. Huang, and H. Wang, "A novel distributed amplifier with high gain, low noise, and high output power in 0.18-μm CMOS technology," *IEEE Trans. Microw. Theory Techn.*, vol. 61, no. 4, pp. 1533–1542, Apr. 2013, doi: 10.1109/TMTT.2013.2247048.
- [48] C. Liu, I. Abdo, C. Wang, H. Sakai, A. Shirane, and K. Okada, "A dual-mode bi-directional CMOS mixer using push-push doubler for 300GHz-band transceivers," in *Proc. IEEE 49th Eur. Solid State Circuits Conf. (ESSCIRC)*, Lisbon, Portugal, Sep. 2023, pp. 69–72, doi: 10.1109/ESSCIRC59616.2023.10268721.
- [49] A. Mirzaei, H. Darabi, J. C. Leete, X. Chen, K. Juan, and A. Yazdi, "Analysis and optimization of current-driven passive mixers in narrowband direct-conversion receivers," *IEEE J. Solid-State Circuits*, vol. 44, no. 10, pp. 2678–2688, Oct. 2009, doi: 10.1109/JSSC.2009.2027937.
- [50] A. Altaf, M. Elahi, S. M. Abbas, J. Yousaf, and E. Almajali, "A D-band waveguide-SIW transition for 6G applications," *J. Electromagn. Eng. Sci.*, vol. 22, no. 4, pp. 419–426, Jul. 2022, doi: 10.26866/jees.2022.4.r.104.
- [51] I. Mohamed and A. Sebak, "Broadband transition of substrate-integrated waveguide-to-air-filled rectangular waveguide," *IEEE Microw. Wireless Compon. Lett.*, vol. 28, no. 11, pp. 966–968, Nov. 2018, doi: 10.1109/LMWC.2018.2871330.
- [52] X. Sun et al., "Causality analyzing for transmission line with surface roughness," in *Proc. IEEE Int. Symp. Electromagn. Compat., Signal Power Integrity (EMC SIPI)*, New Orleans, LA, USA, Jul. 2019, pp. 516–521, doi: 10.1109/ISEMC.2019.8825246.
- [53] M. H. Eissa, N. Maletic, E. Grass, R. Kraemer, D. Kissinger, and A. Malignaggi, "100 Gbps 0.8-m wireless link based on fully integrated 240 GHz IQ transmitter and receiver," in *IEEE MTT-S Int. Microw. Symp. Dig.*, Aug. 2020, pp. 627–630, doi: 10.1109/IMS30576.2020.9224101.
- [54] C. Liu et al., "A 640-Gb/s 4 × 4-MIMO D-band CMOS transceiver chipset," in *Proc. IEEE Symp. VLSI Technol. Circuits*, Jun. 2024, pp. 1–2.



Chenxin Liu (Graduate Student Member, IEEE) received the B.E. degree from Nanjing University of Science and Technology, Nanjing, China, in 2019, and the M.E. degree in electrical and electronic engineering from the Institute of Science Tokyo (formerly Tokyo Institute of Technology), Tokyo, Japan, in 2021, where he is currently pursuing the Ph.D. degree in electrical and electronic engineering.

His current research is about millimeter-wave wireless communication system design.



Chun Wang (Member, IEEE) received the B.E. degree from the University of Electronic Science and Technology of China, Chengdu, China, in 2015, the M.E. degree from Zhejiang University, Hangzhou, China, in 2018, and the Ph.D. degree from the Department of Electrical and Electronic Engineering, Tokyo Institute of Technology, Tokyo, Japan, in 2024.

He is currently a Post-Doctoral Researcher at the Institute of Science Tokyo, Tokyo, focusing on millimeter-wave and terahertz front-end and sys-

tem design. His current research interests include CMOS RF/subterahertz (sub-THz)/THz transceivers, phased-array transceivers, device modeling, and wireless/satellite communication systems.

Dr. Wang was a recipient of the IEEE EDS Japan Joint Chapter Student Award in 2024. He serves as a reviewer for IEEE TRANSACTIONS ON MICROWAVE THEORY AND TECHNIQUES.



Zheng Li (Member, IEEE) received the B.E. and M.E. degrees in microelectronics and solid electronics from Xidian University, Xi'an, China, in 2014 and 2017, respectively, and the Ph.D. degree in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2023.

He has been a Post-Doctoral Researcher with the Institute of Science Tokyo (formerly Tokyo Institute of Technology), Tokyo, since 2023, deeply involved in the research and development of beyond 5G internet of Things system on chip (B5G IoT SoC)

and the construction platform for IoT solutions. His current research interests include radio frequency (RF)/millimeter-wave/analog CMOS phased-array beamformers, 5G area-power-efficient power/low-noise amplifiers, massive multiinput multioutput (MIMO), satellite communication, and 5G/6G high-data-rate mobile systems.

Dr. Li was a recipient of the IEEE Solid-State Circuits Society (SSCS) Student Travel Grant Award in 2023 and the Best Paper Award at the International Conference on Integrated Circuits, Technologies and Applications (ICTA) in 2022. He serves as a reviewer for the IEEE JOURNAL OF SOLID-STATE CIRCUITS, IEEE TRANSACTIONS ON MICROWAVE THEORY AND TECHNIQUES, and IEEE SOLID-STATE CIRCUITS LETTER.



Anyi Tian received the B.E. degree from Tokyo Institute of Technology, Tokyo, Japan, in 2024, where she is currently pursuing the M.E. degree in electrical and electronic engineering.

Her research interests include CMOS RF/subterahertz (sub-THz)/analog transceiver systems, phased arrays, wireless communication, and 6G.



Yudai Yamazaki (Graduate Student Member, IEEE) received the B.E. and M.E. degrees in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2021 and 2023, respectively, where he is currently pursuing the Ph.D. degree in electrical and electronic engineering, working on 5G, beyond 5G and 6G wireless communication system design.

His current research interests include millimeterwave/terahertz CMOS phased-array transceivers.

Mr. Yamazaki was a recipient of the IEEE SSCS Japan Chapter VDEC Design Award in 2023 and the IEICE Best Paper Award in 2023. He serves as a reviewer for the IEEE JOURNAL OF SOLID-STATE CIRCUITS.



Jun Sakamaki received the B.E. and M.E. degrees in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2022 and 2024, respectively.

His research interests include millimeter-wave CMOS wireless transceivers.



Hans Herdian (Member, IEEE) received the B.Sc. degree in electrical engineering from Bandung Institute of Technology, Bandung, Indonesia, in 2016, and the M.S. and Ph.D. degrees in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2018 and 2024, respectively.

His research interests include millimeter- and submillimeter-wave wireless transceiver systems, device modeling, and on-chip passive components modeling and performance enhancement.



Han Nie received the B.E. degree from Zhejiang University, Hangzhou, China, in 2020, and the M.E. degree from Tokyo Institute of Technology, Tokyo, Japan, in 2023. She is currently pursuing the Ph.D. degree in electrical and electronic engineering with the Institute of Science Tokyo, Tokyo.

Her current research interests mainly focus on phased-array subterahertz CMOS wireless communication systems and building block design and intermediate-frequency (IF) distribution networks for D-band systems. LIU et al.: 640-Gb/s 4 × 4-MIMO D-BAND CMOS TRANSCEIVER CHIPSET



Xi Fu (Member, IEEE) received the B.E. degree (Hons.) from Dalian University of Technology, Liaoning, China, in 2017, and the M.E. and Ph.D. degrees from the Department of Electrical and Electronic Engineering, Tokyo Institute of Technology (Tokyo Tech), Tokyo, Japan, in 2019 and 2022, respectively.

He is currently a Post-Doctoral Researcher at Tokyo Institute of Technology. His current research interests are CMOS radio frequency (RF)/millimeter-wave/terahertz/analog transceivers,

phased-array transceivers, mixed-signal systems, 5G/6G mobile systems, device modeling, and satellite communication systems.

Dr. Fu was a recipient of the Japanese Government [the Ministry of Education, Culture, Sports, Science, and Technology of Japan (MEXT)] Scholarship, the IEEE Solid-State Circuits Society (SSCS) Predoctoral Achievement Award in 2023, the RFIC Symposium Best Paper Award in 2019, the IEEE International Solid-State Circuits Conference (ISSCC) Student-Research Preview Poster Award in 2022, and the IEEE SSCS Student Travel Grant Award in 2022. He serves as a reviewer for IEEE TRANSACTIONS ON MICROWAVE THEORY AND TECHNIQUES, IEEE TRANSACTIONS ON VERY LARGE-SCALE INTEGRATION SYSTEMS, IEEE SOLID-STATE CIRCUITS LETTERS, and IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS I & II.



Sena Kato (Graduate Student Member, IEEE) received the B.E. degree in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2023. He is currently pursuing the M.S. degree in electrical and electronic engineering with the Institute of Science Tokyo, Tokyo.

His research interests include CMOS millimeterwave transceiver systems, wireless power transfer systems, satellite communication, and device modeling.



Wenqian Wang (Graduate Student Member, IEEE) was born in Xuancheng, China. He received the B.Eng. degree from the School of Microelectronics, Xidian University, Xi'an, China, in 2019, and the M.Eng. degree from the Institute of Science Tokyo (formerly Tokyo Institute of Technology), Tokyo, Japan, in 2021, where he is currently pursuing the Ph.D. degree.

His current research interests include analog and mixed-signal circuits, as well as frequency synthesizers.



Hongye Huang (Member, IEEE) was born in Guilin, China, in 1994. He received the B.E. degree from the University of Electronic Science and Technology of China, Chengdu, China, in 2016, and the M.E. degree from the Institute of Science Tokyo (formerly Tokyo Institute of Technology), Tokyo, Japan, in 2018, where he is currently pursuing the Ph.D. degree.

He was an Intern with SiTime Japan, Tokyo, in 2022, where he developed digital signal processing for clock synthesis systems. His current research

interests include mixed-signal integrated circuits, frequency synthesizers, and design automation for synthesizable clock circuits.

Mr. Huang was a scholarship recipient of the Watanuki International Scholarship Foundation in fiscal years 2020 and 2021. He currently serves as a reviewer for IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS II: EXPRESS BRIEFS and IEEE ACCESS.



Minzhe Tang (Graduate Student Member, IEEE) received the B.E. degree in microelectronic science and engineering from the Southern University of Science and Technology (SUSTech), Shenzhen, China, in 2020, and the M.E. degree in electrical and electronics engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2022, where he is currently pursuing the Ph.D. degree in electrical and electronic engineering.

His research interests include millimeter-wave CMOS wireless transceivers for 5G mobile.



Dingxin Xu (Member, IEEE) received the B.Eng. degree from the Southern University of Science and Technology, Shenzhen, China, in 2018, and the M.Eng. and Ph.D. degrees from the Institute of Science Tokyo (formerly Tokyo Institute of Technology), Tokyo, Japan, in 2020 and 2024, respectively.

In the summer of 2023, he was an Intern with the RF/Analog Department, Qualcomm, San Diego, CA, USA, where he was designing a frequency synthesizer in an advanced technology node. He was a Post-Doctoral Researcher with the Institute of

Science Tokyo in 2024. After that, he joined Marvell Technology, Singapore, where he is currently a Staff Engineer and focuses on frequency synthesizer design for high-speed wireline communication systems.

Dr. Xu was a recipient of the Tokyo Tech Advanced Human Resource Development Fellowship for Doctoral Students from 2021 to 2023, the ISSCC 2023 Student-Research-Preview (SRP) Poster Award, and the 2024 IEEE SSCS Predoctoral Achievement Award. He serves as a reviewer for the IEEE JOURNAL OF SOLID-STATE CIRCUITS, IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS I: REGULAR PAPERS, and IEEE SOLID-STATE CIRCUITS LETTERS.



Shinsuke Hara (Member, IEEE) received the B.E., M.E., and Ph.D. degrees in physics from Tokyo University of Science, Tokyo, Japan, in 2000, 2002, and 2005, respectively.

In 2013, he joined the National Institute of Information and Communication Technology, Koganei, Japan, as a Researcher. His current research interests include millimeter-wave CMOS circuit design and nanoscale semiconductor devices.



Akifumi Kasamatsu (Member, IEEE) received the B.E., M.E., and Ph.D. degrees in electronics engineering from Sophia University, Tokyo, Japan, in 1991, 1993, and 1997, respectively.

From 1997 to 1999, he was a Research Assistant with Sophia University. From 1999 to 2002, he was with Fujitsu Laboratories Ltd., Atsugi, Japan. Since 2002, he has joined the National Institute of Information and Communications Technology, Koganei, Japan, where he is currently an Executive Researcher and a Principal Investigator of the terahertz wave

electronics project. His current research interests include wireless communication technology, such as wireless transceivers and nanoscale semiconductor devices for millimeter-wave and terahertz wave communications.

Dr. Kasamatsu is a member of the Institute of Electronics, Information and Communication Engineers of Japan, and the Japanese Society of Applied Physics.



Takashi Tomura (Member, IEEE) received the B.E., M.E., and D.E. degrees in electrical and electronic engineering from Tokyo Institute of Technology, Tokyo, Japan, in 2008, 2011, and 2014, respectively. From 2014 to 2017, he worked at Mitsubishi Electric Corporation, Tokyo, and was engaged in the research and development of aperture antennas for satellite communications and radar systems. From 2017 to 2019, he was a Specially Appointed Assistant Professor at Tokyo Institute of Technology, where he is currently an Assistant Professor. His

research interests include electromagnetic analysis, aperture, reflect array, and waveguide slot array antennas.

Dr. Tomura is a member of the Institute of Electronics, Information, and Communication Engineers (IEICE). He was a Research Fellow of the Japan Society for the Promotion of Science (JSPS) in 2013. He received the Best Student Award from Ericsson Japan in 2012, the IEEE Antennas and Propagation Society (AP-S) Tokyo Chapter Young Engineer Award in 2015, the Young Researcher Award from IEICE Technical Committee on Antennas and Propagation in 2018, and the IEEE Microwave Theory and Techniques Society (MTT-S) Japan Chapter Young Engineer Award in 2022.



Hiroyuki Sakai (Senior Member, IEEE) received the B.S. and M.S. degrees in electrical engineering from Osaka University, Osaka, Japan, in 1984 and 1986, respectively, and the Ph.D. degree from the University of Fukui, Fukui, Japan, in 2020.

In 1986, he joined the Semiconductor Research Center, Matsushita Electric Industrial Company Ltd., Osaka, and engaged in research and development of high-speed GaAs digital ICs, GaAs RF ICs for very compact cellular phones. In 1993, he started to research and develop millimeter-wave devices and

their ICs, which resulted in the invention of a new millimeter-wave IC concept named millimeter-wave flip-chip IC (MFIC). From 1998 to 2000, he visited Stanford University, Stanford, CA, USA, as a Visiting Scholar, and expanded his research subjects to new Si-based RF devices and their integration technologies. From 2012 to 2017, he continued his research on mm-wave ICs based on GaAs, GaN, Si-BiCMOS, and CMOS large scale integration (LSI) technologies at some laboratories of Panasonic Corporation, Osaka. In 2020, he joined Tokyo Institute of Technology, Tokyo, Japan, as a Creative Manager of the Open Innovation Platform, where he is currently a Specially Appointed Professor of electrical and electronic engineering.

Prof. Sakai was a member of the Technical Program Committee of the IEEE International Solid-State Circuit Conference (ISSCC) from 2002 to 2008. He served as a Secretary of the IEEE Electron Devices Society Kansai Chapter from 2002 to 2003.



Atsushi Shirane (Member, IEEE) received the B.E. degree in electrical and electronic engineering and the M.E. and Ph.D. degrees in electronics and applied physics from Tokyo Institute of Technology, Tokyo, Japan, in 2010, 2012, and 2015, respectively. From 2015 to 2017, he was with Toshiba Corporation, Kawasaki, Japan, where he developed an 802.11ax Wireless LAN RF transceiver. From 2017 to 2018, he was with Nidec Corporation, Kawasaki, where he researched intelligent motors with wireless communication. He is currently an

Associate Professor with the Laboratory for Future Interdisciplinary Research of Science and Technology, Institute of Integrated Research, Institute of Science Tokyo, Tokyo. His current research interests include RF CMOS transceivers for IoT, 5G, satellite communication, and wireless power transfer.

Dr. Shirane has been a member of the Technical Program Committee for International Solid-State Circuits Conference Student Research Preview since 2019. He is a member of the IEEE Solid-State Circuits Society, the Institute of Electronics, Information and Communication Engineers (IEICE), and the Japan Institute of Electronics Packaging (JIEP). He was a recipient of the Docomo Mobile Science Awards for Excellence in Advanced Technology in 2024.



Kenichi Okada (Fellow, IEEE) received the B.E., M.E., and Ph.D. degrees in communications and computer engineering from Kyoto University, Kyoto, Japan, in 1998, 2000, and 2003, respectively.

In 2003, he joined Tokyo Institute of Technology, Tokyo, Japan, as an Assistant Professor, where he is currently a Professor of electrical and electronic engineering at the Institute of Science Tokyo, Tokyo. He has authored or co-authored more than 500 journal and conference papers. His current research interests include millimeter-wave and tera-

hertz CMOS wireless transceivers for 20/28/39/60/77/79/100/300 GHz for 5G, WiGig, satellite and future wireless systems, digital phase-locked loop (PLL), synthesizable PLL, atomic clock, and ultralow-power wireless transceivers for Bluetooth low-energy, and sub-GHz applications.

Prof. Okada is a member of the Institute of Electrical and Electronics Engineers (IEEE), the Institute of Electronics, Information and Communication Engineers (IEICE), the Information Processing Society of Japan (IPSJ), and the Japan Society of Applied Physics (JSAP). He is/was a member of the Technical Program Committees of the IEEE International Solid-State Circuits Conference (ISSCC), VLSI Circuits Symposium, European Solid-State Circuits Conference (ESSCIRC), Radio Frequency Integrated Circuits Symposium (RFIC), and Asian Solid-State Circuits Conference (A-SSCC). From 2000 to 2003, he was a Research Fellow of the Japan Society for the Promotion of Science at Kyoto University. He was a recipient or co-recipient of the Ericsson Young Scientist Award in 2004, the A-SSCC Outstanding Design Award in 2006 and 2011, the ASP-DAC Special Feature Award in 2011, the Best Design Award in 2014 and 2015, the MEXT Young Scientists' Prize in 2011, the JSPS Prize in 2014, the Suematsu Yasuharu Award in 2015, the MEXT Prizes for Science and Technology in 2017, the RFIT Best Paper Award in 2017, the IEICE Best Paper Award in 2018, the RFIC Symposium Best Student Paper Award in 2019, the IEICE Achievement Award in 2019, the DOCOMO Mobile Science Award in 2019, the IEEE/ACM ASP-DAC Prolific Author Award in 2020, the Kenjiro Takayanagi Achievement Award in 2020, the KDDI Foundation Award in 2020, the IEEE CICC Best Paper Award in 2020, the IEEE ISSCC Author-Recognition Award in 2023, and more than 50 other international and domestic awards. He is/was a Guest Editor and an Associate Editor of IEEE JOURNAL OF SOLID-STATE CIRCUITS, an Associate Editor of IEEE TRANSACTIONS ON MICROWAVE THEORY AND TECHNIQUES, and a Distinguished Lecturer of the IEEE Solid-State Circuits Society (SSCS).



Kazuaki Kunihiro (Member, IEEE) received the B.S. and M.S. degrees in applied physics from Tokyo Institute of Technology, Tokyo, Japan, in 1988 and 1990, respectively, and the D.E. degree in quantum engineering from Nagoya University, Nagoya, Japan, in 2004.

From 1990 to 2023, he was with NEC Corporation, Kawasaki, Japan, where he engaged in research and development on device modeling of GaAs and GaN FETs, high-efficiency power amplifiers for mobile base stations, millimeter-wave

(mmWave)/subterahertz (sub-THz) transmission systems for mobile backhaul, and multimode/multiband transceiver ICs for software-defined radio systems. From 1995 to 1996, he was a Visiting Researcher at the Technical University of Berlin, Berlin, Germany, where he worked on nonlinear physics in III–V compound semiconductor devices. Since July 2023, he has been serving as a Specially Appointed Professor with the Department of Electrical and Electronic Engineering, School of Engineering, Institute of Science Tokyo, Tokyo. His current research interests include mmWave/sub-THz phasedarray antennas and multiinput multioutput (MIMO) systems for B5G/6G and satellite communications.

Dr. Kunihiro is a member of the Microwave Theory and Techniques Society, the Solid-State Circuits Society, and the Institute of Electronics, Information and Communication Engineers (IEICE). He served as a Technical Program Committee (TPC) Member for the IEEE Compound Semiconductor IC Symposium from 2013 to 2015.